\begin{table}
\caption{Parameters
for the exponential disc fitting.} \label{expdiscparams}
%\centerline
{
\begin{tabular}{lccccccc}
\hline
  \hline
\noalign{\smallskip}
Parameter [Unit] & NGC~2366 & ESO~059-G001 & ESO~215-G?009 & NGC~4861 &
  NGC~5408 & IC~5152\\
& (1) & (2) & (3) & (4) & (5) & (2)\\
\hline
\noalign{\smallskip}
Band & \emph{V} & \emph{H} & \emph{R} & \emph{R} & \emph{J} & \emph{H}\\
$\mu_0~[{\rm mag~arcsec}^{-2}]$ & 22.82 & 20.23 & 23.16 & 22.95 & 18.29 & 17.65\\
$h$ [kpc] & 1.59 & 0.81 & 2.16 & 0.88 & 0.33 & 0.36\\
$n$ & 1 & 1.33 & 1 & 1 & 1 & 0.96\\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
$M/L$ & 1 & 0.8 & 2 & 0.3 & <0.1 & 0.2\\
\hline 
\end{tabular}}
\par
References: Photometry by (1) \citet{Hunter2001}; (2)
\citet{Kirby2008}; (3) \citet{Warren2006}; (4)
\citet{GildePaz2005}; (5) \citet{Noeske2003}.
\end{table}